Author's response to reviews

Title: Recurrent Posterior Circulation Strokes Caused by the Anomalous Occipital Bony Process in a Young Patient

Authors:

Seung-Hoon Song (<20100111@kuh.ac.kr>)
Hong Gee Roh (<hgroh@kuh.ac.kr>)
Hahn Young Kim (<hykimmd@gmail.com>)
Jin Woo Choi (<cardijin@gmail.com>)
Won-Jin Moon (<mdmoonwj@gmail.com>)
Woo Jin Choe (<s1nsman@chol.com>)
Ileok Jung (<kahram77@naver.com>)

Version: 2
Date: 5 August 2014

Author's response to reviews: see over
August 5, 2014

Dr. Timothy Shipley
Executive Editor, BMC neurology

Dear Dr. Shipley

My coauthors and I submit the manuscript, Recurrent Posterior Circulation Strokes Caused by the Anomalous Occipital Bony Process in a Young Patient for publication in BMC Neurology.

Stroke associated with occipital bony process is thought to be a rare condition. It could be easily overlooked and underestimated because this anatomical variation is very trivial. However, the consequence of this small process could be serious. I am assure this report will grab readers’ attention and probably alert them.

Moreover, to our knowledge, this is the first report of a endovascular vertebral artery occlusion for prevent recurrent strokes caused by the abnormal occipital bony process.

We certify that all authors have read and approved submission of the manuscript, and material in the manuscript has not been published and is not being considered for publication elsewhere in whole or in part in any language.

Thank you for your consideration of our paper. We look forward to hearing from you.

Sincerely yours,

Hong Gee Roh, MD, PhD
Department of Radiology
Konkuk University Medical Center, Konkuk University School of Medicine
120-1 Neungdong-ro, Gwangjin-gu, Seoul, Korea
Tel: +82-2-2030-5545 Fax: +82-2-2030-5549
E-mail address: hgroh@kuh.ac.kr